Author's response to reviews

Title: Mood color choice helps to predict response to hypnotherapy in patients with irritable bowel syndrome

Authors:

Helen R Carruthers (helen.carruthers@uhsm.nhs.uk)
Julie Morris (julie.morris@manchester.ac.uk)
Nicholas Tarrier (nicholas.tarrier@manchester.ac.uk)
Peter J Whorwell (peter.whorwell@manchester.ac.uk)

Version: 3 Date: 23 September 2010

Author's response to reviews:

Dear Sir/Madam,

We have made the following changes as requested:

1) The author list on the paper has been changed to match those given in the submission.

2) The country has been added to the affiliations.

3) The visible vertical lines have been removed from the tables.

4) The position of the references in the text has been changed as requested.

Best wishes,

Peter Whorwell